Author’s response to reviews

Title: The oncogenic effects of HES1 on salivary adenoid cystic carcinoma cell growth and metastasis

Authors:

Yu Huang (770483176@qq.com)
Huan Gan (595332465@qq.com)
Jian Xie (463904646@qq.com)
Lin She (mykim2002@163.com)
Yong Zhao (297392511@qq.com)
Can Ding (dingdentist@163.com)
Hua Su (sudoctor2005@126.com)
Dali Zheng (dalizheng@health.usf.edu; dalizheng@hotmail.com)
Guang Lu (fjlyg63@163.com)

Version: 2 Date: 22 Mar 2018

Author’s response to reviews:

Response to Editors

Editor Comments:

Please perform the following revisions to your manuscript:

1. Please add all authors’ email addresses before the abstract.

We added all authors’ email addresses.
2. On uploading your revisions, please remove all tracking/highlighting and instead upload a single clean version of the manuscript.

We accepted all the changes and will upload the final one. Thanks!

3. Under 'Ethical approval and consent to participate' (within declarations) please add your ethical approval reference number.

We added the IRB reference number to the “Cell culture and clinical samples” section in the Methods and to the ” Ethics approval and consent to participate” section.

4. Under Funding (within declarations) - please add the phrase: 'The funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript'.

We added the phrase to the end of Funding section.